Author's response to reviews

Title: Nitrous Oxide Does Not Increase the Risk of Cancer Recurrence after Colorectal Surgery: A Randomized, Blinded Study

Authors:

Edith Fleishmann (Edith.fleischmann@meduniwien.ac.at)
Corinna Marschalek (Friedrich.herbst@meduniwien.ac.at)
Katja Schlemitz (Katja_schlemitz@gmx.at)
Jarrod Dalton (DaltonJ@ccf.org)
Thomas Gruenberger (Thomas.gruenberger@meduniwien.ac.at)
Herbst Friedrich (Friedrich.herbst@meduniwien.ac.at)
Andrea Kurz (AK@OR.org)
Daniel Sessler (DS@OR.org)

Version: 2 Date: 3 December 2008

Author's response to reviews:

Dear Editor,

With regard to manuscript MA 1614674353227923

We have added the clinical trial number and website information on the first page of the application. Also, in the abstract on that page, the clinical trial number and website have been added at the end of the abstract.

A new copy of the whole manuscript is being uploaded that includes the new abstract with the clinical trial information.

Regards,

DS